Author's response to reviews

Title: Expression of estrogen receptor beta in the breast carcinoma of BRCA1 mutation carriers

Authors:

Maria M Litwiniuk (litwiniuk@skrzynka.pl)
Krzysztof K Roznowski (krzysztof.roznowski@oncology.am.poznan.pl)
Violetta V Filas (vfilas@wp.pl)
Dariusz D Godlewski (godlewski.open@wp.pl)
Małgorzata M Stawicka (gosiastawicka.open@wp.pl)
Remigiusz R Kaleta (remigiusz.kaleta@pharmacia.com)
Jan J Breborowicz (janbreborowicz@wp.pl)

Version: 4 Date: 20 March 2008

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

Thank you very much for your mail dated 14. Mar 2008. In the enclosure is a manuscript which has been corrected according to your leads: updated authors list – deleted email addresses, added initials of forenames, changed affiliations list, sector headings, added conclusions, changed tables according to your leads.

Looking forward to hearing you,

Yours sincerely,

Maria Litwiniuk